# STUDY PROTOCOL Gene-Environment Interactions in Major Mental Disorders in the Czech Republic

This article was published in the following Dove Press journal: Neuropsychiatric Disease and Treatment

## Klara Latalova 🕞 Omar Sery (D<sup>2,3</sup> Kristyna Hosakova 🗗 Ladislav Hosak<sup>4</sup>

<sup>1</sup>Department of Psychiatry, Palacky University Olomouc, School of Medicine and University Hospital Olomouc, Olomouc, Czech Republic; <sup>2</sup>Department of Biochemistry, Faculty of Science, Masaryk University, Brno, Czech Republic; <sup>3</sup>Laboratory of Neurobiology and Pathological Physiology, Institute of Animal Physiology and Genetics, Academy of Sciences of the Czech Republic, Brno, Czech Republic; <sup>4</sup>Department of Psychiatry, Charles University, School of Medicine in Hradec Kralove and University Hospital Hradec Kralove, Hradec Kralove, Czech Republic Background: Mental disorders affect about one-third of the human population, are typically chronic and significantly decrease the quality of life. Presently, the treatment of mental illnesses is far from adequate with a substantial proportion of the patients being pharmacoresistant and suffering from relapses. One of the reasons for this complicated situation is that we do not precisely know about the causes of mental disorders, so their treatment cannot be causal. The etiology of a mental disorder is typically based on a combination of molecular (genetic) and environmental factors.

Aim: The aim of the project is to discover the gene-environment interactions (GxE) in a wide spectrum of mental disorders.

Methods: The design of our study is innovative in the sense that we intend to study large groups of associated mental disorders as a whole instead of in isolation. This would enable us to map out the possible environmental causal factors in detail in relation to their character, magnitude and timing. The project also allows a study of genetics (including epigenetics and microbiomes) as well as the environment simultaneously. We plan on involving three study groups: the first group are patients suffering from schizophrenia or a mood disorder such as major depression, recurrent depressive disorder and bipolar affective disorder; the second group of patients have anxiety disorders; and the third group are healthy volunteers from the general population who are genetically unrelated. All of the study subjects will undergo the following assessments: a psychiatric examination, the identification of stressful life events with the aid of a questionnaire, the examination of their reaction to stress, genetic and epigenetic (microRNA) assessments and the analysis of oral and gut microbiome.

**Conclusion:** We expect that some of the genetic as well as environmental factors in the studied mental disorders are shared, while some others are specific. We also expect that the GxE (gene-environment interaction) in schizophrenic and affective disorders will be different from the GxE in anxiety disorders and that the GxE in the studied mental disorders will differ generally from the GxE in healthy volunteers. Our results can help in the prevention and individualized treatment of a range of mental disorders.

Keywords: schizophrenia, mood disorders, anxiety disorders, gene, environment, GxE interactions

## Plain Language Summary

- Mental disorders affect about one third of the human population and are typically lifelong. They significantly impair the subjects' quality of life and may lead to suicide.
- Their treatment result is not satisfactory, partly because our knowledge of their etiology is limited, so the therapy is more symptomatic than causal.
- The etiology of almost all mental disorders is based on the interaction of genes and the environment.

1147

Neuropsychiatric Disease and Treatment 2020:16 1147-1156

© 2020 Latalova et al. This work is published and licensed by Dove Medical Press Limited. The full terms of this license are available at https://www.dovepress.com/terms. work you hereby accept the Terms. Non-commercial uses of the work are permitted without any further permission from Dove Medical Press Limited, provided the work is properly attributed. For permission for commercial use of this work, please see paragraphs 4.2 and 5 of our Terms (https://www.dovepress.com/terms.php).

Correspondence: Ladislav Hosak Tel +420 495 832 360 Fax +42 495 83 3 041 Email hosak@lfhk.cuni.cz



- Previous research was mostly aimed at either genetic or environmental factors in isolation and not on their mutual interactions, which led to a bias.
- We propose carrying out a complex study on the interaction between genes and the environment in major mental disorders (psychosis, mood disorders and anxiety disorders).
- We will provide a questionnaire to the patients about stressful life events, we will carry out an examination of the patients' reaction to stress, do genetic and epigenetic (microRNA) studies and analyse the oral and gut microbiomes of the patients and we will perform the same assessments in healthy subjects who are genetically unrelated.
- We expect that the results of our project may help in the treatment and prevention of major mental disorders.
- The proposed project due to its focus and comprehensiveness represents a priority in the field of psychiatry not only in the Czech Republic but throughout the rest of Central Europe as well.

# Background

A lifetime prevalence of mental disorders is estimated to be in the range 12.0–47.4%.<sup>1</sup> Once a mental disorder develops, eg dementia, substance addiction, psychosis, mood or anxiety disorders, it usually continues until the end of the patient 's life. Other mental disorders, like personality disorders or intellectual disability, are lifelong.<sup>2</sup> Apart from suicide, a mental disorder does not usually lead directly to the death of the patient, but it significantly decreases his or her quality of life, increases the level of disability in the population and places a considerable economic burden on society.<sup>3</sup>

In spite of the attained progress that has been made in treating mental illness, the current treatment regimens of mental disorders are still far from satisfactory. We are able to treat most mental disorders and improve the clinical state of the patient, but a full and continuous remission is rather rare. One of the reasons is that we do not know the exact etiology of mental disorders very well, so the treatment is more likely to be symptomatic rather than causative.<sup>4</sup>

Most mental disorders require interactions between the genes and the environment to develop.<sup>5-10</sup> If the genetic or environmental factors of a mental disorder are studied separately, the research results would inevitably be biased. The solution to this issue is to study both the genetic and the environmental factors simultaneously, in order to find connections between the two. Some of these findings were already published and replicated, such as the influence of the expression of the MAOA (monoamine oxidase A) gene and the

degree of violence experienced in childhood on antisocial behavior in adulthood or the influence of stress on the development of depression depending on the BDNF (brain-derived neurotrophic factor) Val66Met polymorphism.<sup>11,12</sup>

A recent approach on testing the statistical interactions between the genetic and environmental factors in the complex etiology of mental disorders is the genome-wide environment interaction study (GWEIS).<sup>5</sup> This way, the genome of study subjects as well as the significant environmental factors able to influence the development of a mental disorder (exposome) are assessed in a comprehensive way. Similarly to the evaluation of genetic risk factors (polygenic risk scores), the environmental risk factors are also appraised via the usage of an omnibus test (polyenvironmental risk scores). These studies where a lot of genetic and environmental factors are both assessed together would have more validity than research aimed only at genes or the environment. It is useful to study psychiatric diagnoses in a cumulative way, as groups of diseases rather than isolated clinical entities, because: (a) the diagnosis of a mentally ill subject frequently changes throughout his or her life; (b) comorbidities within mental disorders are pervasive, and the risk persists over time;  $^{13}$  (c) not all recent psychiatric diagnoses have a sufficient validity (eg schizoaffective disorder or mixed anxiety-depressive disorder); (d) many detected genetic polymorphisms represent a shared heritability in common disorders of the brain;<sup>14</sup> and (e) many environmental factors are nonspecific: eg either seriously stressful life events may induce major depression in some subjects but episodes of schizophrenia in other ones. In addition to this, different environmental factors may contribute to the development of the same disease in predisposed individuals; eg in schizophrenia there were found to be correlations with the season of the patient's birth, the occurrence of obstetric complications, a history of abuse or trauma during childhood, the use of cannabis as well as exposure to stressful events.<sup>15</sup> Even if the specificities of the individual environmental factors as related to one psychiatric diagnosis are low, more specificity may be attained by a unique combination of these E (environmental) factors. An excellent review of how to "embrace" the reality of the development of mental illness due to the complex interplay of genetic and environmental factors was presented by Uher et Zwicker.<sup>16</sup> According to the authors, the GxE interactions should be assessed in a prospective, longitudinal way throughout the whole life of the individual (developmental context), with a specific focus on the first two decades of life. This type of

research can be performed without any a priori hypotheses (hypothesis-free design), which was already common and fully accepted in the GWAS (genome-wide association studies) in psychiatry in the past. In most mental disorders, the consecutive and complex interactions of GxGxExE (a complex interplay) are present rather than a simple causation by GxE. The timing of these interactions is significant, eg because while cannabis abuse may easily induce schizophrenia in adolescence in a predisposed boy or girl, it rarely leads to the same disease in older subjects. The genetic background of the patient, usually polygenic, sensitizes humans to the influence of extrinsic factors (geneenvironment interaction; GxE). Childhood trauma in many cases, results in an increased sensitivity to stress thus leading to anxiety and depression in adulthood.<sup>17</sup> It is also necessary to assess the pregnancy of the patient's mother, because many deleterious factors already start working in utero. The occurrence of seriously stressful life events should also be looked into concerning the patient's ancestors, because of the intergenerational transmission of trauma via epigenetic mechanisms.<sup>18</sup> Protective environmental factors which have for the most part been neglected in research; eg a happy childhood, should also be evaluated in a GWEIS. Eventually, statistically significant GxEs ought to be independently replicated and be able to serve as a foundation for future research into their underlying causal biological mechanisms.16

MicroRNAs (miRNAs) are small, highly conserved non-coding RNA (ribonucleic acid) molecules, containing about 22 nucleotides that are involved in the regulation of gene expression. Recent studies show that miRNAs are significant biomarkers of schizophrenia, bipolar disorder, major depression and anxiety disorders.<sup>19–22</sup> That is why we consider the miRNome analysis in our study subjects as an integral and complementary part of genetic research.

Microbiota are one of the key regulators of gut-brain function and have led to the appreciation of the importance of the microbiota-gut-brain axis. This axis starts to be significant in investigations of the biological background of neurodevelopmental and neurodegenerative mental disorders.<sup>23</sup> In a mouse model, it appears that the manipulation of the intestinal microbiome alters the expression of genes in the brain.<sup>24</sup> Some studies that were done on psychiatric patients suggest that the use of antibiotics is associated with an increased risk of psychotic episodes. Oro-pharyngeal microbiota in schizophrenia were found to be different between the cases and the controls.<sup>25</sup> Mental disorders are generally associated with a reduced microbial diversity and show differences in the human microbiome compared to the controls. In some reports, specific microbial taxa have been associated with physical health as well as with symptoms of depression, psychosis, etc.<sup>26</sup> It has been shown that changes in the gut microbiome following the administration of quetiapine in patients with bipolar disorder, could be seen as a marker of prediction of the outcome of treatment.<sup>27</sup> Studies also suggest increased intestinal inflammation and permeability, which may be among the principal mechanisms by which microbial dysbiosis influences the overall physiological function of the patient.<sup>26</sup> In addition to that, microbiome may correlate with some miRNA levels, so miRNAs may be one of the possible links between the brain and the intestine.<sup>28</sup>

## Study Aims

The aim of our GWEIS study is to conduct a survey of the gene–environment interactions in the etiopathogenesis of selected mental disorders that are serious in nature and to compare them with unrelated healthy volunteers. The basic design of the proposed study is cross-sectional, but we plan to follow-up the available study subjects prospectively. We intend to measure the exposome from a life-long perspective. Because stress is one of the most important factors that can trigger an episode of a mental disorder, the study is also centred on the assessment of the subject's reactivity to stress.

We plan to compare the miRNA profiles among our study groups using unsupervised hierarchical clustering analysis. We expect that the differences in the miRNAs' expression will be related not only to the psychiatric disorders themselves, but also to the lifestyle of the patient, their microbiome and their reactivity to stress. We will analyse the microbiome from samples taken from oral and stool swabs and compare the data among the study groups as well as to assess the microbiomes in relation to the miRNAs levels, the lifestyle of the studied subjects as well as their diet.

# Theoretical Hypotheses

Even if our attitude is skeptical and theory-free, it is logical to assume that

 the study groups will be different from each other in the presence of individual environmental risk factors and environmental protective factors as well as the polyenviromic risk score and the polyenviromic protective score;

- the study groups will differ from each other in the presence of individual genetic polymorphisms as well as the polygenic risk score;
- the study groups will be different from each other regarding the GxE (product of the polyenviromic risk score and polygenic risk score);
- the study groups will be different from each other in terms of their miRNAs expression and their microbiomes.

# **Methods**

# The Participants and Study Setting

Three groups of study subjects will be evaluated:

- Patients suffering from schizophrenia or a mood disorder (major depression, recurrent depressive disorder, bipolar affective disorder)
- Patients with anxiety disorders
- Unrelated volunteers from the general population who are mentally healthy.

The study subjects, who are all volunteers, will be recruited among the patients and staff (healthy volunteers) at the departments of psychiatry and at the university hospitals (healthy volunteers) in Hradec Kralove and Olomouc. The psychiatric patients who visited the hospital for the first time as well as the patients who are undergoing continuous psychiatric treatment will be enrolled.

The patients who will be recruited at the psychiatry department will already have been diagnosed with a mental disorder. There is no way to know if the healthy volunteers will develop a mental disorder later in life, which could be a limitation of the study. Mentally healthy volunteers who have a history of psychiatric illness will not be included in our study.

# **Evaluations**

#### Psychiatric Examination

Each volunteer will be examined by a licensed psychiatrist (K.L., L.H.) as of the possible recent development of psychopathology and it's diagnosis according to the DSM-5 (Diagnostic and Statistical Manual of Mental Disorders, 5<sup>th</sup> Edition) classification.<sup>29</sup>

#### The Questionnaire of Stressful Life Events

We adapted the original questionnaire of stressful life events applied in the EU-GEI (European network of National Schizophrenia Networks Studying Gene–Environment Interactions) study on psychosis.<sup>30,31</sup> This questionnaire covers the subject's whole life, including possible adverse agents and situations in utero and also if there were any major stressful life events in the subject's parents and grandparents.

For the purpose of our GxE study, we will look into the presence of environmental factors from the questionnaire in every subject, whether they are present or not.

Of interest to us are the environmental risk factors for schizophrenia in which their effect size is known:<sup>32,33</sup>

- Whether the subject was born in the winter or spring.
- Upbringing in an urban area.
- History of cannabis abuse.
- Advanced age of the father when the subject was conceived.
- Presence of obstetric or perinatal complications.
- History of physical abuse, sexual abuse or neglect.
- Death of a parent when the subject was a child.
- Being a member of an ethnic minority.

For mood and anxiety disorders, we are interested in the following environmental risk factors in which their effect size is known: $^{34-41}$ 

- Alcohol abuse.
- Smoking.
- Serious chronic somatic disease.
- Traumatic brain injury.
- Obesity.
- A low level of physical activity.
- The presence of any job related stress.
- Shift work.
- Lack of sleep.
- Seriously stressful events in adulthood.
- Migration.
- Ethnic minority status.
- Living a solitary way of life.

(Remark: Some of the risk factors for mood and anxiety disorders are the same as the risk factors for schizophrenia, eg stressful life events in childhood or cannabis abuse, and were already mentioned in the Padmanabhan's list of risk factors.<sup>32</sup>)

Protective factors for mood and anxiety disorders in which their effect size is known:<sup>42–44</sup>

- A Mediterranean diet.
- Physical activity.
- Social support.

#### The Examination of Reactivity to Stress

We will apply the Virtual Reality Version of the Trier Social Stress Test (TSST-VR).<sup>45</sup> Cortisol in the subject's saliva will be measured shortly before the test, immediately after the test, and 10, 20, 30 and 40 minutes after the test. Within the same time period, the heart rate will be continuously assessed using pulse oximetry. This will be supplemented by the subject's statement on how much stress is being perceived by him/herself. Within one week before the TSST-VR, we will measure the circadian patterns of salivary cortisol, because these patterns are altered in major mental disorders, eg schizophrenia.<sup>46</sup> We will collect a sample of saliva 3 times during a 24 hour period. The first sample will be collected immediately after waking up, the second one 30 minutes later and the third one hour before the patients sleeps. It is found that the salivary cortisol values follow the same pattern as serum cortisol itself, and have a high correlation with some parameters showing disturbances in the hypothalamus-pituitary-adrenal axis.47

#### Genetic Assessments

Blood samples of all study subjects will be genotyped at the Department of Biochemistry, at the Faculty of Science of Masaryk University in Brno, in the Czech Republic. For the analysis of DNA (deoxyribonucleic acid) polymorphisms, the most up-to-date method will be used - the targeted re-sequencing using the NGS (next-generation sequencing) system NextSeq from Illumina. For a targeted re-sequencing of highly polymorphic and/or gene function sites of selected genes, the NimbleGen SeqCap Target Enrichment method enabling the specific enrichment of gene regions of up to 96 samples will be used. Individual samples for sequencing analysis will be multiplexed in order to obtain about 100x depth of coverage, which is generally accepted as adequate in the detection of SNPs (single-nucleotide polymorphisms). The data will be analysed according to the workflow provided by the manufacturer for the evaluation of the NimbleGen SeqCap EZ Target Enrichment Data consisting of the assessment of the quality of sequencing reads, read filtering, mapping against the human genome, duplicate removals, the assessment of coverage statistics, variant calling and variant filtering. The final data will be exported to a file for statistical analyses. The technology described above is routinely used in our laboratories.48,49

We will focus on the genes relevant to the mental disorders under study: ADRA2A, AKT1, BDNF, CCKAR, CHI3L1, CHRNA7, CLDN5, CNR1, COMT, CRFR1, CTXN3, DISC1, DPYSL2, DRD3, DTNBP1, EGF, GABRA1, GABRB2, GAD1, GRIK3, GRM3, HTR2A, HTR3A, IL10, MAOA, MAOB, MCHR1, MLC1, NEUROG1, NOS1, NOS1AP, NOTCH4, NPY, NRG1, NR3C1, OLIG2, OPMR1, PDE4B, PLA2G4A, PLXNA2, PPP3CC, PRODH, RELN, RGS4, RTN4, SLC6A4, SNAP25, SRR, SYN2, TNF, TP53, UFD1L, XBP1, YWHAE, ACE, TNF alpha, TGF beta, MTHFR, IL1, IL2, IL6, CRP, GSK3 and others according to the lists published by Schmidt-Kastner et al, Roussos et Haroutunian, Hosak and Jiang et al and according to our previous studies.<sup>50–57</sup>

#### Epigenetic (miRNA) Assessment

We will use next-generation sequencing to analyse the miRNAs in all samples.

The serum sample will be stored in a stabilizing solution. The collected samples will be used for miRNA isolation and the RNA will be stored at -70°C until its use. For the preparation of miRNA sequencing libraries, we will use the NEBNext<sup>®</sup> Small RNA Library Prep Set for Illumina (New England Biolabs, MA, USA). The sequencing libraries will be subsequently loaded in the NextSeq 500 sequencing device (Illumina, CA, USA). The online databases such as the TargetScan, microT-CDS and Tarbase will be used to identify the target genes of the differentially expressed miRNAs.

#### **Microbiome Analysis**

Oral and gut microbiota will be collected from the study subjects by obtaining oral smears and faecal specimens. The determination of bacteria will be performed by the metagenomic sequencing method by using 16s rRNA. Samples of stool will be elaborated according to the instructions for the QIAamp PowerFecal DNA Kit (Qiagen, Germany). The collected samples of buccal swabs will be used for gDNA/total RNA isolation using the AllPrep DNA/RNA/miRNA Universal Kit (Qiagen) according to the manufacturer's instructions and the DNA/RNA will be stored at -20°C/-70°C until use. Isolated DNA will be used as a template in PCRs (polymerase chain reactions) with the 16S modified 515F-806R primer pair flanking the V4 hypervariable region, including the linker, pad, barcode and the P5 and P7 Illumina adapters.<sup>58</sup> Amplification will be done according to an already introduced protocol using the Platinum SuperFi Mastermix (Thermo) in a volume of 25 ul when amplification products will be purified by using the Agencourt<sup>®</sup> AM Pure XP beads. Consequently, the purified PCR samples will be checked and quantified by using the Qubit, Fragment Analyser (Advanced Analytical) and the Kapa Library Quantification Kit (Roche). The final library will be subjected to NGS using a MiniSeq<sup>®</sup> High Output Kit (2 x 150 paired end sequencing) on a MiniSeq<sup>®</sup> sequencer (Illumina, USA). The Mothur software package will be used for sequence analysis. OTU detection, taxonomic assignment, and phylogenetic analysis.<sup>59</sup> We will calculate the species richness (alpha diversity) by using the Chao, Shannon and Simpson indexes. We will analyse the statistically significant changes in beta diversity by utilizing the weighted variant of UniFrac. The analysis of the principal components, using the Jaccard index, will be carried out with a Phyloseq package, to compare the bacterial populations among different samples and to define the core microbiome in our samples.<sup>60,61</sup> All of the data analysis will be done according to the Human Microbiome Project (NIH; National Institutes of Health) pipeline to allow data deposition and comparison with the already analysed samples. Real-time PCR with primers flanking the V3/V4 hypervariable region will be carried out to quantify total bacteria.

# Data Monitoring

The data will be continuously collected and entered into an Excel table by the researchers. All databases will be organized anonymously, according to the unique ID numbers which will be assigned to the individual participants at the beginning of the study. The data will not be accessible on the internet. The access to the data will be restricted only to the research team.

# Statistical Analysis

All of the statistical analyses will be carried out using the R software (R Core Team).<sup>62</sup>

The polyenviromic risk score (PERS) will be computed for each participant as recommended by Padmanabhan et al, similarly to the calculations of the polygenic risk scores.<sup>32</sup> We will binarize environmental risk factors, and apply an additive model. The log of the odds ratio for each environmental risk factor will be multiplied by either "1" (risk factor present in the individual) or "0" (risk factor absent in the individual). These products will be added together and the sum will be divided by the total number of assessed risk factors. The association of selected environmental risk factors with the relevant mental disorder will be tested using the Fisher's exact test.

We will calculate the polygenic risk score (PRS) as suggested by Dudbridge.<sup>63</sup> Each of the single nucleotide polymorphisms (SNPs) will be tested for an association with the relevant mental disorder (based on the case/control status of an individual) using the Fisher's exact test. Only the SNPs with a frequency higher than 5% will be included in the analysis. SNPs with a frequency lower than 5% will be analysed as being potential mutations. We will determine whether a mutation can affect gene function (eg a change of the amino acid of a protein). In the event that a mutation can affect the gene function, we will proceed by analysing the occurrence of this mutation in relation to the mental disorders in the families being studied, in the same manner as we have done in the work on the genetics of oligodontia.48 Then, all of the SNPs with P values exceeding the selected threshold and being approximately in linkage equilibrium with the others, will be used to calculate the PRS for each participant using the logistic regression model. The polygenic risk score uses an additive model to quantify an individual's genetic loading for a disorder, as influenced by the presence of multiple risk alleles. To calculate an individual's polygenic risk score, the log of the odds ratio for each risk allele is multiplied by the number of risk alleles (0, 1 or 2) in the subject. This is done for all SNPs that are included in the score. These products are added together and divided by the total number of single nucleotide polymorphisms included in the score.32

To test the joint effects of the selected environmental risk factors and polygenic risks, the PRS will be binarized according to the procedure suggested by Guloksuz et al.<sup>64</sup> According to this procedure, a participant is considered to have significant polygenic risk when exceeding the third quartile of the values of PRS in the control group. Based on this, four groups will be delimited for each environmental factor: (1) participants with the polygenic risk of a relevant mental disorder, but without exposition to the selected environmental factor (eg cannabis abuse); (2) participants exposed to the environmental factors, but without polygenic risk; (3) participants exposed both to the environmental factors and polygenic risk; (4) reference exposed neither to the environmental factors nor polygenic risk.

The interactions between each environmental factor and the polygenic risk of the relevant mental disorder will be assessed by comparing the values or the frequency variations of the environmental factors between these four groups. Multivariate logistic regression will be used to test the differences that control the influence of the patient's sex and age as well as the environmental factors that will be assessed. By using this method, the information that will be obtained will indicate whether the environmental factors act alone or through the interaction with polygenic risks. Similarly, the differences in the expression of miRNA between these groups and the diversity in the microbial species (Chao index) will be tested by using the non-parametric Kruskal–Wallis test to explore their relationship with the environmental and genetic risk factors.

We will gain additional information by running the analysis for specific haplotypes using the PERS method.

According to the literature, study groups > 1000 individuals in GxE psychiatric research are sufficiently large enough to ensure a sufficient statistical power to detect an effect of a given size with a sufficient degree of confidence (the power level of 0.80; the significance criterion with a value of 0.05).<sup>64</sup>

## Ethical Issues

The study will be performed in agreement with the Declaration of Helsinki as amended in 2018.<sup>65</sup> The study was approved by the Ethics Committee of the University Hospital of Hradec Kralove (Reference number: 201903 S10P) and the Ethics Committee at the University Hospital of Olomouc (Reference number: 25/19). All participants will voluntarily sign the written informed consent form.

#### Timetable

Preparation phase: November 2019 – December 2019. Data collection phase: January 2020 - December 2022, Data analysis and publishing: January 2023 – December 2023.

## Discussion

The Czech Republic is a Central European country. Its area of 78,866 km<sup>2</sup> is comparable in size to Hungary, Austria or Ireland. The number of inhabitants is about 10.6 million and the unemployment rate is equal to 2.8%. The proportion of people living below the poverty threshold is very low as compared to other European countries because of the long tradition of egalitarian policies of the Czech Republic. About ten per cent of the population is represented by immigrants, mostly coming from Slovakia, Ukraine, Vietnam or Russia. Only one third of the population declare themselves to be religious, with the prevailing religion being that of the Roman Catholic faith (83.4%). The average life expectancy in the Czech

Republic is 78.3 years and the fertility-rate runs on average at 1.57 deliveries per woman of child bearing age. The most recent gross domestic product per capita totals to 35.010 USD. The Czech Republic is an industriallydeveloped country with a long tradition of strength in the sectors of machinery production, automotive manufacturing and chemical technologies. Some things that make the Czech Republic stand out in a positive light is the high employment rate among women and a quality healthcare system that is highly affordable and easily accessible. It is also worth noting that the Czech Republic has one of the lowest perinatal mortality-rates in the world, a tolerant attitude to sexual minorities and a high degree of vaccinations in the general population against contagious diseases.<sup>66</sup> On the other hand, the Czech Republic has one of the higher rates of alcohol consumption among European countries.<sup>67</sup> As far as we know, the suggested study will be the first GWEIS in the Czech Republic as well as in Central Europe.

The advantage of our project is that it is a comprehensive assessment of gene–environment interactions in major mental disorders. Standard GxE evaluations are supplemented with a virtual test of a reactivity to stress with the detection of miRNAs and microbiomes, which demonstrates recent knowledge on the important factors in gene–environment issues.<sup>68,69</sup> Our study is innovative in the sense that we intend to look at the combinations of various environmental factors while also considering their timing, so as not to study them in isolation from each other.

In order to obtain results that are both robust and meaningful, we will make use of the findings of our cross-sectional study to formulate hypotheses which we plan to test in a consecutive study on GxE with a longitudinal design.

With the extension of the study, we intend to perform an analysis of the endophenotypes that are relevant to schizophrenia (eg prepulse inhibition, mismatch negativity, oculomotor antisaccade, letter-number sequencing and continuous performance tests), major depression (eg neuroticism, morning cortisol deviation, frontal asymmetry of cortical electrical activity and biases of attention and memory), bipolar disorder (eg deficits in verbal memory, sustained attention and executive function or abnormal structural and resting state brain neuroimaging findings) and anxiety disorders (eg aberrant amygdala responsiveness to anxiety-related stimuli).<sup>70–73</sup>

The proposed project is limited by the fact that we will only look for the statistically significant gene–environment interactions and their neurobiological explanation may be

1153

sought after in further studies. The suggested examination of the reactivity to stress only reflects psychosocial stress, but this type of stress is prominent in the induction of an episode of a mental disorder.<sup>4</sup> We are aware of the fact that the TSST-VR only measures responses to acute stressors and not chronic psychosocial stressors, which would be more relevant to the longitudinal aspect of the environmental factors we intend to monitor. That is the reason why we will interpret the TSST-VR results with caution, strictly in association with the data obtained from the questionnaire of stressful life events in every subject. As far as we know, there is currently no test for assessing chronic psychosocial stress in humans. Another limitation of our study is that the population being surveyed, in comparison to other countries like the United States for example, is not as genetically and culturally diverse because of the relative homogeneity of the Czech population as a whole. Thus the GxE interactions highlighted by the results may only be relevant to the Caucasian populations of Central European descent, but not necessarily to other populations.

# Conclusion

The aim of our project is to initiate pioneering GxE research in the field of psychiatry in the Czech Republic. After we collect the pilot results and publish them, we intend to apply for a research grant. The results of the project may become applicable in the study of neurobiology of major mental disorders, and possibly in their treatment and prevention. We would welcome and greatly appreciate any comments from the readers that could serve in improving our work.

# **Abbreviations**

BDNF, brain-derived neurotrophic factor; DNA, deoxyribonucleic acid; DSM-5, Diagnostic and Statistical Manual of Mental Disorders, 5th Edition; E, environmental; EU-GEI, European network of National Schizophrenia Networks Studying Gene–Environment Interactions; GxE, gene–environment interaction; GWAS, genome-wide association study; GWEIS, genome-wide environment interaction study; MAOA, monoamine oxidase A; miRNA, microRNA; NGS, next-generation sequencing; NIH, National Institutes of Health; PCR, polymerase chain reaction; PERS, polyenviromic risk score; PRS, polygenic risk score; RNA, ribonucleic acid; SNP, single-nucleotide polymorphism; TTST-VR, the Virtual Reality Version of the Trier Social Stress Test.

#### Acknowledgments

This work was sponsored by the Czech Health Research Council of the Ministry of Health of the Czech Republic. The reference number of the research grant from the Czech Health Research Council is AZV MZ ČR 16-27243A. The sponsor had no involvement in any stage of preparation of this paper.

The authors thank Dr. Tomas Zeman, Ph.D. for his help with the statistical parts of the manuscript and Cyrus Rasti, M.D. for his language revision of the paper.

# Disclosure

The authors report no conflicts of interest in this work.

## References

- Kessler RC, Angermeyer M, Antohony JC, et al. Lifetime prevalence and age-of-onset distributions of mental disorders in the World Health Organization's World Mental Health Survey Initiative. *World Psychiatry*. 2007;6(3):168–176.
- 2. Hosak L, Hrdlicka M, editors. *Psychiatry and Pedopsychiatry*. 1st ed. Prague: Karolinum Press; 2016.
- Institute for Health Metrics and Evaluation (IHME). Findings from the Global Burden of Disease Study 2017. Seattle: IHME; 2018. Available from: http://www.healthdata.org/sites/default/files/files/pol icy\_report/2019/GBD\_2017\_Booklet.pdf. Accessed March 6, 2019.
- Sadock BJ, Sadock VA, Ruiz P, Kaplan HI, editors. Kaplan & Sadock 's Comprehensive Textbook of Psychiatry. 9th ed. Philadelphia: Wolters Kluwer Health/Lippincott Williams & Wilkins; 2009.
- Zwicker A, Denovan-Wright EM, Uher R. Gene-environment interplay in the etiology of psychosis. *Psychol Med.* 2018;48 (12):1925–1936. doi:10.1017/S003329171700383X
- Ludwig B, Dwivedi Y. Dissecting bipolar disorder complexity through epigenomic approach. *Mol Psychiatry*. 2016;21(11):1490–1498. doi:10. 1038/mp.2016.123
- Van der Auwera S, Peyrot WJ, Milaneschi Y, et al. Genome-wide gene-environment interaction in depression: a systematic evaluation of candidate genes: the childhood trauma working-group of PGC-MDD. *Am J Med Genet B Neuropsychiatr Genet.* 2018;177 (1):40–49. doi:10.1002/ajmg.b.32593
- Sharma S, Ressler KJ. Genomic updates in understanding PTSD. *Prog Neuropsychopharmacol Biol Psychiatry*. 2019;90:197–203. doi:10.1016/j.pnpbp.2018.11.010
- Schiele MA, Domschke K. Epigenetics at the crossroads between genes, environment and resilience in anxiety disorders. *Genes Brain Behav.* 2018;17(3):e12423. doi:10.1111/gbb.12423
- Baker JH, Schaumberg K, Munn-Chernoff MA. Genetics of Anorexia Nervosa. *Curr Psychiatry Rep.* 2017;19(11):84. doi:10.1007/s11920-017-0842-2
- Ouellet-Morin I, Cote SM, Vitaro F, et al. Effects of the MAOA gene and levels of exposure to violence on antisocial outcomes. *Br J Psychiatry*. 2016;208(1):42–48. doi:10.1192/bjp.bp.114.162081
- Zhao M, Chen L, Yang J, et al. BDNF Val66Met polymorphism, life stress and depression: a meta-analysis of gene-environment interaction. *J Affect Disord*. 2018;227:226–235. doi:10.1016/j.jad.2017.10.024
- Plana-Ripoll O, Bocker Pedersen C, Holtz Y, et al. Exploring comorbidity within mental disorders among a Danish National Population. JAMA Psychiatry. 2019;76(3):259–270. doi:10.1001/jamapsychiatry.2018.3658

- Anttila V, Bulik-Sullivan B, Finucane HK, The Brainstorm Consortium. Analysis of shared heritability in common disorders of the brain. *Science*. 2018;360(6395):eaap8757. doi:10.1126/science. aap8757
- Modinos G, Lyegbe C, Prata D, et al. Molecular genetic gene-environment studies using candidate genes in schizophrenia: a systematic review. *Schizophrenia Res.* 2013;150(2–3):356–365. doi:10.1016/j.schres.2013.09.010
- Uher R, Zwicker A. Etiology in psychiatry: embracing the reality of poly-gene-environmental causation of mental illness. *World Psychiatry*. 2017;16(2):121–129. doi:10.1002/wps.20436
- Huh HJ, Kim KH, Lee HK, Chae JH. The relationship between childhood trauma and the severity of adulthood depression and anxiety symptoms in a clinical sample: the mediating role of cognitive emotion regulation strategies. J Affect Disord. 2017;213:44–50. doi:10.1016/j.jad.2017.02.009
- Costa DL, Yetter N, DeSomer H. Intergenerational transmission of paternal trauma among US Civil War ex-POWs. *Proc Natl Acad Sci* U S A. 2018;115(44):11215–11220. doi:10.1073/pnas.1803630115
- He K, Guo C, Guo M, et al. Identification of serum microRNAs as diagnostic biomarkers for schizophrenia. *Hereditas*. 2019;156:23. doi:10.1186/s41065-019-0099-3
- Fries GR, Carvalho AF, Quevedo J. The miRNome of bipolar disorder. J Affect Disord. 2018;233:110–116. doi:10.1016/j. jad.2017.09.025
- 21. Ferrua CP, Giorgi R, da Rosa LC, et al. MicroRNAs expressed in depression and their associated pathways: a systematic review and a bioinformatics analysis. *J Chem Neuroanat*. 2019;100:101650. doi:10.1016/j.jchemneu.2019.101650
- Murphy CP, Singewald N. Role of MicroRNAs in Anxiety and Anxiety-Related Disorders. *Curr Top Behav Neurosci*. Epub 2019. 1–35.
- Cryan JF, O'Riordan KJ, Cowan CSM, et al. The microbiota-gutbrain axis. *Physiol Rev.* 2019;99(4):1877–2013.
- 24. Chu C, Murdock MH, Jing D, et al. The microbiota regulate neuronal function and fear extinction learning. *Nature*. 2019;574(7779):543–548. doi:10.1038/s41586-019-1644-y
- Dickerson F, Severance E, Yolken R. The microbiome, immunity, and schizophrenia and bipolar disorder. *Brain Behav Immun*. 2017;62:46–52. doi:10.1016/j.bbi.2016.12.010
- Nguyen TT, Kosciolek T, Eyler LT, Knight R, Jeste DV. Overview and systematic review of studies of microbiome in schizophrenia and bipolar disorder. J Psychiatr Res. 2018;99:50–61. doi:10.1016/j. jpsychires.2018.01.013
- 27. Lu Q, Lai J, Lu H, et al. Gut microbiota in bipolar depression and its relationship to brain function: an advanced exploration. *Front Psychiatry*. 2019;10:784. doi:10.3389/fpsyt.2019.00784
- Mansour MA, Sabbah NA, Mansour SA, Ibrahim AM. MicroRNA-199b expression level and coliform count in irritable bowel syndrome. *IUBMB Life*. 2016;68(5):335–342. doi:10.1002/iub.1495
- American Psychiatric Association. *Diagnostic and Statistical Manual* of *Mental Disorders*. 5th(DSM-5) ed. Washington, D.C.: American Psychiatric Association Publishing; 2013.
- European Network of National Networks studying Gene-Environment Interactions in Schizophrenia (EU-GEI). Identifying gene-environment interactions in schizophrenia: contemporary challenges for integrated, large-scale investigations. *Schizophr Bull*. 2014;40(4):729–736. doi:10.1093/schbul/sbu069
- Quattrone D, Di Forti M, Gayer-Anderson C, et al. Transdiagnostic dimensions of psychopathology at first episode psychosis: findings from the multinational EU-GEI study. *Psychol Med.* 2019;49 (8):1378–1391. doi:10.1017/S0033291718002131
- 32. Padmanabhan JL, Shah JL, Tandon N, Keshavan MS. The "polyenviromic risk score": aggregating environmental risk factors predicts conversion to psychosis in familial high-risk subjects. *Schizophr Res.* 2017;181:17–22. doi:10.1016/j.schres.2016.10.014

- Jongsma HE, Turner C, Kirkbride JB, Jones PB. International incidence of psychotic disorders, 2002–17: a systematic review and meta-analysis. *Lancet Public Health*. 2019;4(5):e229–e244. doi:10.1016/S2468-2667(19)30056-8
- 34. Boden JM, Fergusson DM. Alcohol and depression. *Addiction*. 2011;106(5):906–914. doi:10.1111/j.1360-0443.2010.03351.x
- 35. Chang-Quan H, Xue-Mei Z, Bi-Rong D, Zhen-Chan L, Ji-Rong Y, Qing-Xiu L. Health status and risk for depression among the elderly: a meta-analysis of published literature. *Age Ageing*. 2010;39 (1):23–30. doi:10.1093/ageing/afp187
- 36. Köhler CA, Evangelou E, Stubbs B, et al. Mapping risk factors for depression across the lifespan: an umbrella review of evidence from meta-analyses and Mendelian randomization studies. *J Psychiatr Res.* 2018;103:189–207. doi:10.1016/j.jpsychires.2018.05.020
- 37. Luger TM, Suls J, Vander Weg MW. How robust is the association between smoking and depression in adults? A meta-analysis using linear mixed-effects models. *Addict Behav.* 2014;39(10):1418–1429. doi:10.1016/j.addbeh.2014.05.011
- Marangoni C, Hernandez M, Faedda GL. The role of environmental exposures as risk factors for bipolar disorder: a systematic review of longitudinal studies. *J Affect Disord*. 2016;193:165–174. doi:10.1016/j. jad.2015.12.055
- Perry DC, Sturm VE, Peterson MJ, et al. Association of traumatic brain injury with subsequent neurological and psychiatric disease: a meta-analysis. J Neurosurg. 2016;124(2):511–526. doi:10.3171/ 2015.2.JNS14503
- 40. Stubbs B, Koyanagi A, Hallgren M, et al. Physical activity and anxiety: a perspective from the World Health Survey. J Affect Disord. 2017;208:545–552. doi:10.1016/j.jad.2016.10.028
- Swinnen SG, Selten JP. Mood disorders and migration: meta-analysis. Br J Psychiatry. 2007;190:6–10. doi:10.1192/bjp.bp.105.020800
- Gariepy G, Nitka D, Schmitz N. The association between obesity and anxiety disorders in the population: a systematic review and meta-analysis. *Int J Obes.* 2010;34(3):407–419. doi:10.1038/ ijo.2009.252
- 43. Lai JS, Hiles S, Bisquera A, Hure AJ, McEvoy M, Attia J. A systematic review and meta-analysis of dietary patterns and depression in community-dwelling adults. *Am J Clin Nutr.* 2014;99 (1):181–197. doi:10.3945/ajcn.113.069880
- 44. Schuch FB, Vancampfort D, Firth J, et al. Physical activity and incident depression: a meta-analysis of prospective cohort studies. *Am J Psychiatry*. 2018;175(7):631–648. doi:10.1176/appi.ajp.2018.17111194
- 45. Wallergard M, Jonsson P, Osterberg K, Johansson G. A virtual reality version of the trier social stress test: a pilot study. *Presence-Teleop Virt Env.* 2011;20(4):325–336. doi:10.1162/PRES\_a\_00052
- 46. Coulon N, Brailly-Tabard S, Walter M, Tordjman S. Altered circadian patterns of salivary cortisol in individuals with schizophrenia: a critical literature review. *J Physiol Paris*. 2016;110(4Pt B):439–447. doi:10.1016/j.jphysparis.2017.05.002
- Asok M, Natarajan PM, Saeed A, et al. A study on comparison of salivary cortisol circadian rhythm in periodontal diseases with external stressors and clinical parameters. *Biomed Pharmacol J.* 2016;9 (2):679–688. doi:10.13005/bpj/990
- 48. Bonczek O, Bielik P, Krejci P, et al. Next generation sequencing reveals a novel nonsense mutation in MSX1 gene related to oligodontia. *PLoS One.* 2018;13(9):e0202989. doi:10.1371/journal. pone.0202989
- Zapletalova M, Kasparovska J, Krizova L, Kasparovsky T, Sery O, Lochman J. Bacterial community dynamics in a rumen fluid bioreactor during in-vitro cultivation. *J Biotechnol.* 2016;234:43–49. doi:10. 1016/j.jbiotec.2016.07.013
- Schmidt-Kastner R, van Os J, Esquivel G, Steinbusch HW, Rutten BP. An environmental analysis of genes associated with schizophrenia: hypoxia and vascular factors as interacting elements in the neurodevelopmental model. *Mol Psychiatry*. 2012;17:1194–1205. doi:10.1038/ mp.2011.183

- Roussos P, Haroutunian V. Schizophrenia: susceptibility genes and oligodendroglial and myelin related abnormalities. *Front Cell Neurosci.* 2014;8:5. doi:10.3389/fncel.2014.00005
- Hosak L. New findings in the genetics of schizophrenia. World J Psychiatry. 2013;3(3):57–61. doi:10.5498/wjp.v3.i3.57
- Jiang S, Postovit L, Cattaneo A, Binder EB, Aitchison KJ. Epigenetic modifications in stress response genes associated with childhood trauma. *Front Psychiatry.* 2019;10:808. doi:10.3389/fpsyt.2019. 00808
- 54. Sery O, Lochman J, Povova J, Janout V, Plesnik J, Balcar VJ. Association between 5q23.2-located polymorphism of CTXN3 gene (Cortexin 3) and schizophrenia in European-Caucasian males; implications for the aetiology of schizophrenia. *Behav Brain Funct*. 2015;11:10. doi:10.1186/s12993-015-0057-9
- Lochman J, Plesnik J, Janout V, et al. Interactive effect of MTHFR and ADRA2A gene polymorphisms on pathogenesis of schizophrenia. *Neuro Endocrinol Lett.* 2013;34(8):792–797.
- 56. Sery O, Prikryl R, Castulík L, Stastny F. A118G polymorphism of OPRM1 gene is associated with schizophrenia. J Mol Neurosci. 2010;41(1):219–222. doi:10.1007/s12031-010-9327-z
- 57. Lochman J, Balcar VJ, Stastny F, Sery O. Preliminary evidence for association between schizophrenia and polymorphisms in the regulatory regions of the ADRA2A, DRD3 and SNAP-25 genes. *Psychiatry Res.* 2013;205(1–2):7–12. doi:10.1016/j.psychres.2012.08.003
- 58. Caporaso JG, Lauber CL, Walters WA, et al. Global patterns of 16S rRNA diversity at a depth of millions of sequences per sample. *Proc Natl Acad Sci USA*. 2011;108(Suppl1):4516–4522. doi:10.1073/ pnas.1000080107
- 59. Kozich JJ, Westcott SL, Baxter NT, Highlander SK, Schloss PD. Development of a dual-index sequencing strategy and curation pipeline for analyzing amplicon sequence data on the MiSeq Illumina sequencing platform. *Appl Environ Microbiol.* 2013;79(17):5112–5120. doi:10. 1128/AEM.01043-13
- Lozupone C, Knight R. UniFrac: a new phylogenetic method for comparing microbial communities. *Appl Environ Microbiol*. 2005;71(12):8228–8235. doi:10.1128/AEM.71.12.8228-8235.2005
- McMurdie PJ, Holmes S, Watson M. Phyloseq: an R package for reproducible interactive analysis and graphics of microbiome census data. *PLoS One.* 2013;8(4):e61217. doi:10.1371/journal.pone.006 1217

- 62. R Core Team. A Language and Environment for Statistical Computing. Vienna: R foundation for statistical computing; 2018.
- Dudbridge F, Wray NR. Power and predictive accuracy of polygenic risk scores. *PLoS Genet*. 2013;9(3):e1003348. doi:10.1371/journal. pgen.1003348
- 64. Guloksuz S, Pries LK, Delespaul P, et al. Examining the independent and joint effects of molecular genetic liability and environmental exposures in schizophrenia: results from the EUGEI study. *World Psychiatry*. 2019;18(2):173–182. doi:10.1002/wps.20629
- 65. WMA Declaration of Helsinki ethical principles for medical research involving human subjects [homepage on the Internet]. The World Medical Association; 2019. Available from: https://www.wma.net/poli cies-post/wma-declaration-of-helsinki-ethical-principles-for-medicalresearch-involving-human-subjects/. Accessed March 7, 2019.
- Eurostat [homepage on the Internet]. The Czech Statistical Office; 2018. Available from: https://www.czso.cz/csu/czso/eurostat. Accessed March 6, 2019.
- 67. The Czech Statistical Office [homepage on the Internet]. The Czech Statistical Office; 2018. Available from: https://www.czso.cz/csu/czso/cr\_od\_roku\_1989\_alkohol. Accessed March 8, 2019.
- No authors listed. Links between gut microbes and depression strengthened. *Nature*. 2019;566:7. doi:10.1038/d41586-019-00483-5
- 69. Lucas G. Gut thinking: the gut microbiome and mental health beyond the head. *Microb Ecol Health Dis*. 2018;29(2):1548250. doi:10.1080/ 16512235.2018.1548250
- Greenwood TA, Shutes-David A, Tsuang DW, et al. Endophenotypes in schizophrenia: digging deeper to identify genetic mechanisms. *J Psychiatr Brain Sci.* 2019;4(2). doi:10.20900/jpbs.20190005
- Goldstein BL, Klein DN. A review of selected candidate endophenotypes for depression. *Clin Psychol Rev.* 2014;34(5):417–427.
- 72. Miskowiak KW, Kjaerstad HL, Meluken I, et al. The search for neuroimaging and cognitive endophenotypes: a critical systematic review of studies involving unaffected first-degree relatives of individuals with bipolar disorder. *Neurosci Biobehav Rev.* 2017;73:1–22. doi:10.1016/j.neubiorev.2016.12.011
- Domschke K, Dannlowski U. Imaging genetics of anxiety disorders. *Neuroimage*. 2010;53(3):822–831. doi:10.1016/j.neuroimage.2009.11. 042

#### Neuropsychiatric Disease and Treatment

#### **Dove**press

Publish your work in this journal

Neuropsychiatric Disease and Treatment is an international, peerreviewed journal of clinical therapeutics and pharmacology focusing on concise rapid reporting of clinical or pre-clinical studies on a range of neuropsychiatric and neurological disorders. This journal is indexed on PubMed Central, the 'PsycINFO' database and CAS, and is the official journal of The International Neuropsychiatric Association (INA). The manuscript management system is completely online and includes a very quick and fair peer-review system, which is all easy to use. Visit http://www.dovepress.com/testimonials.php to read real quotes from published authors.

Submit your manuscript here: https://www.dovepress.com/neuropsychiatric-disease-and-treatment-journal